Author’s response to reviews

Title: Perceptions of unmet healthcare needs: What Do Punjabi and Chinese-speaking immigrants think? A qualitative study

Authors:

   Emily G Marshall (Emily.Marshall@dal.ca)
   Sabrina T Wong (sabrina.wong@nursing.ubc.ca)
   Jeannie L. Haggerty (Jean.Lois.Haggerty@USherbrooke.ca)
   Jean-Frederic Levesque (jflevesq@santepub-mtl.qc.ca)

Version: 6 Date: 28 January 2010

Author's response to reviews:

I am attaching the final proofed version. thanks,
SW